Author’s response to reviews

Title: How to Incorporate Broader Outcomes in Economic Evaluations of Public Health Interventions: A qualitative analysis based on semistructured experts interviews

Authors:

Ghislaine APG van Mastrigt PhD (g.vanmastrigt@maastrichtuniversity.nl)
Aggie TG Paulus PhD (a.paulus@maastrichtuniversity.nl)
Marie-Jeanne Aarts PhD (mj.aarts@maastrichtuniversity.nl)
Silvia MAA Evers Prof. (s.evers@maastrichtuniversity.nl)
Adrienne FG Alayli-Goebbels PhD (adrienne.alayli@oahpp.ca)

Version: 2 Date: 7 November 2014

Author’s response to reviews:

Maastricht, 7th November 2014.

Dear reviewers,

Thank you very much for given us the opportunity to revise our manuscript entitled: “How to Incorporate Broader Outcomes in Economic Evaluations of Public Health Interventions: A qualitative analysis based on semistructured experts interviews.”

As requested we checked whether we could improve the reporting of our study by using the “RATS guidelines”.

Our study was not discussed with an Ethics Committee, as according to the Dutch law no ethical approval is required for this type of study.

We hope that this manuscript complies with the requirements of BMC Public Health and that it will be considered for publication.

Sincerely yours,

Ghislaine van Mastrigt on behalf of the authors